Author's response to reviews

Title: Cytokeratin 8/18 expression indicates a poor prognosis in squamous cell carcinomas of the oral cavity

Authors:

Thomas Fillies (fillies@uni-muenster.de)
Richard Werkmeister (Rwerkmeis@aol.com)
Jens Packeisen (jpackeisen@pathoweb.de)
Burkhard Brandt (brandt@uni-muenster.de)
Philippe Morin (p.morin@katharinenhospital.de)
Dieter Weingart (weingart.mkg@katharinenhospital.de)
Ulrich Joos (joos@uni-muenster.de)
Horst Buerger (burgerh@uni-muenster.de)

Version: 3 Date: 20 November 2005

Author's response to reviews: see over
MS: 1233974546691840 - Cytokeratin 8/18 expression indicates a poor prognosis in squamous cell carcinomas of the oral cavity

Comments to revision:

We appreciate the comments of the reviewers and include the suggestions in the revised paper. We regret the obvious mistake in the Material and Method section and added the respective staining technique.